Author's response to reviews

Title: Overexpression of SERBP1 (Plasminogen activator inhibitor 1 RNA binding protein) in human breast cancer is correlated with favourable prognosis

Authors:

Nuran Bektas Serce (bnuran@hotmail.com)
Andreas Boesl (boeselchen14@web.de)
Irina Klaman (irina.klaman@signature-diagnostics.de)
Sonja v Serenyi (svonserenyi@ukaachen.de)
Erik Noetzel (enoetzel@ukaachen.de)
Michael F Press (press@usc.edu)
Arno Dimmler (info@pathologie-karlsruhe.de)
Arndt Hartmann (arndt.hartmann@uk-erlangen.de)
Jalid Sehouli (Sehouli@aol.com)
Ruth Knuechel (rknuechel-clarke@ukaachen.de)
Matthias W Beckmann (fk-direktion@uk-erlangen.de)
Peter A Fasching (peter.fasching@uk-erlangen.de)
Edgar Dahl (edahl@ukaachen.de)

Version: 3 Date: 3 December 2012

Author's response to reviews: see over
Overexpression of SERBP1 (Plasminogen activator inhibitor 1 RNA binding protein) in human breast cancer is correlated with favourable prognosis

Responses to reviewers' comments:

Dear Professor Lacson and dear Professor Helou,
we would like to thank you for the valuable comments which helped us to further improve the manuscript. Please find enclosed our response letter. The revised parts of the text have been marked (red colour).

Referee 1’s (Hiroko Yamashita) comments:
Major Compulsory Revisions
None.

Referee 2’s (Yiannis Drosos) comments:
Major Compulsory Revisions
Comment 1:
Discretionary revision:
1. The phrase “…SERBP1 is a putative tumour suppressor gene…” is an overstatement and should be rephrased.
Response 1:
We appreciate the reviewer’s comments very much. We have removed this phrase (see page 20) and replaced it by the sentence: “In summary, SERBP1 is a potential prognostic marker in human breast carcinoma.”